Reviewer 05755618 comments:
The author comments on the article “Endoscopic mucosal ablation - an alternative treatment for colonic polyps: Three case reports” by Mendoza Ladd A et al. They suggested the disadvantage of EMA is that it does not obtain a surgical specimen for pathological analysis. They advised that EMA should not be used for larger lesions. Although the indication of EMA should be determined for a further large number of studies, the authors presented a helpful comment in this article.

Authors reply: Since no changes were suggested by the reviewer we kindly thank him/her for the kind review.

Reviewer 05722857 comments:
I fully agree with the authors of the Commenting letter to the editor. The original paper they comment on is interesting, however all the comments raised by the Letter to the editor are very important. Ablation of large colonic polyps by ABC is in my eyes out of necessity virtue and other options such as EMR or ESD should be considered first. Only correct that No 1 and 2 citations are the same (original) article so the citations should be renumbered.

Authors reply: Thanks for your comment. I have revised the references.